Author's response to reviews

Title: Pancreatitis and myocarditis followed by pulmonary hemorrhage, a rare presentation of leptospirosis- A case report and literature survey

Authors:

Nuwan Ranawaka (nuwan.rk@gmail.com)
Vijayabala Jeevagan (jeevagany@yahoo.com)
Panduka Karunanayake (pandukaru@mail.com)
Saroj Jayasinghe (sarojoffice@yahoo.com)

Version: 2 Date: 23 August 2012

Author's response to reviews: see over
23rd Aug 2012

Dr. Tonilynn Manibo
Editor
Journal Editorial Office
BioMed Central.

Dear Dr. Manibo,

Sending the revised manuscript of “Pancreatitis and myocarditis followed by pulmonary hemorrhage, a rare presentation of leptospirosis- A case report and literature survey”:

MS: 1428781764786316

Thank you very much for reviewing the article and for the suggestions to improve the manuscript. We made the following change as you requested.

1) Removal of the texts from image 2.

We would be grateful to you, if you could review the revised manuscript.

Thank you for your kind assistance.

Kind Regards,

Nuwan Ranawaka
University Medical Unit,
National Hospital of Sri Lanka.